Author's response to reviews

**Title:** Biomonitoring of bisphenol A concentrations in maternal and umbilical cord blood in regard to birth outcomes and adipokine expression: a birth cohort study in Taiwan

**Authors:**

Wei-Chun Chou (d9712871@oz.nthu.edu.tw)
Jyh-Larng Chen (jlchen@mail.ypu.edu.tw)
Chung-Fen Lin (chungfen1119@yahoo.com.tw)
Yi-Chun Chen (cyc@mail.cjcu.edu.tw)
Feng-Cheng Shih (fengcheng@mail.ypu.edu.tw)
Chun-Yu Chuang (cychuang@mx.nthu.edu.tw)

**Version:** 5  **Date:** 21 October 2011

**Author's response to reviews:** see over
Dear Dr. Grandjean and Dr. Herz:

We have revised the manuscript according to the suggestions. We simplified the description in the Results text, and removed the causal language in the Discussion. In fact, last revision has been edited by an agency but still not ready for publication. Thus, we contact other agency to proof read the document. Hopefully, this revision can reach the standard for publication.

Best regards,

Chun-Yu Chuang  
Associate professor  
Department of Biomedicine Engineering and Environmental Sciences  
National Tsing Hua University, Taiwan  
E-mail: cychuang@mx.nthu.edu.tw